Author’s response to reviews

Title: A lung cancer patient with deep vein thrombosis: A case report and literature review

Authors:

Yungu Chen (chenyg@shsmu.edu.cn)
Yiu Tsang (tsangys0325@gmail.com)
Xiaoxia Chou (chouxiaoxia@renji.com)
Jiong Hu (hujiongzhi@163.com)
Qing Xia (jingtianxq@163.com)

Version: 4 Date: 10 Mar 2019

Author’s response to reviews:

Dear editors:

Thank you for your precious comments on our manuscript, which really helped us a lot in terms of the improvement of the manuscript’s quality. We have made some modifications according to your comments point by point.

Comment 1. Discussion and Conclusions

In order to be in line with journal requirements, please combine your "Discussion" and "Conclusions" sections into one "Discussion and Conclusions" section.

Response 1. Thank you for your comment. The “Conclusions” section has been integrated into the “Discussion” section, together as “Discussion and Conclusions” section now.

Comment 2. Overlap

Thank you for the changes you have made to your manuscript to reduce the overlap. There are still some areas of high overlap in the Background and Discussion sections. Please see the attached document for examples of areas with high textual overlap, please note this document does not contain all areas of concern.

While we understand that you may wish to express some of the same ideas contained in these publications, please be aware that we cannot condone the use of text from previously published work.
Please re-phrase these sections to minimise overlap.

Response 2. Thank you for your comment. We looked through our manuscript thoroughly and made changes in those texts of potential overlap. A native English speaker who was professional in language editing was consulted to help avoid any change in the original meanings of these areas while rephrasing them. We will also upload a copy of the manuscript with tracked changes as a supplemental material in hopes of highlighting the changes we have made and reducing the amount of your work.

Comment 3. Clean Manuscript

Thank you for uploading a clean version of your manuscript, please again upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please also ensure that all figures, tables, and supplementary/additional files are cited in the text.

Response 3. Thank you for your comment. The manuscript and all other materials will be uploaded as your request.

We sincerely thank you again for your work and patience.

Kind regards

Yungu Chen